Point-by-point response to the Reviewers’ comments

Reviewer 1. Thank you very much for the kind words of appreciation of our paper. Concerning the comments:

1. In the conclusion section we added a few sentences about resolution of Wells’ syndrome as you suggested.

Reviewer 2. Thank you very much for the kind words of appreciation of our paper. Concerning the comments:

1. Thank you for the correction, it is leg not arm on the Figure 2.